\documentclass[11pt,twoside,a4paper,fleqn]{book} 
%\documentclass[11pt,twoside,a4paper,fleqn,draft]{book} 

\input{common}

% External documents for cross references:
\externaldocument[D-]{arts_developer}
\externaldocument[T-]{arts_theory}

%===   Start of report   ===================================================
\begin{document}


%--- Title page
%
% To suppress hyperref warning about duplicate page labels:
\renewcommand{\thepage}{title \arabic{page}} 

\thispagestyle{plain}
\begin{center}
  \vspace*{1cm}
  {\Huge \bf ARTS User Guide\\}
  \vspace*{1cm}
  {\large edited by \\}
  \vspace*{1cm}
  {\Large \bf Patrick Eriksson and Stefan Buehler }\\
   \vspace*{2cm}
   {\large \today\\
    ARTS Version \input{auto_version}
   }
\end{center}
\vspace*{\fill}
{\normalsize \bf
  \noindent
  The content and usage of ARTS are not only described by this
  document. An overview of ARTS documentation and help features are
  given in Section~\ref{sec:concept:doc}. For continuous reports on
  changes of the source code and this user guide, subscribe to the
  ARTS developers mailing list at \url{http://www.sat.ltu.se/arts/support/}.

  We welcome gladly comments and reports on errors in the software or the
  document. Send then an e-mail to: \verb|patrick.eriksson(at)chalmers.se| or
  \verb|sbuehler(at)ltu.se|.

  If you use data generated by ARTS in a scientific publication, then please
  mention this and cite the most appropriate of the ARTS publications. The
  relevant publications are summarised at
  \url{http://www.sat.ltu.se/arts/docs/}. }

\newpage                          
\thispagestyle{empty}
\vspace*{\fill}
\noindent
\begin{code}
Copyright (C) 2000-2011
Stefan Buehler <sbuehler (at) ltu.se>
Patrick Eriksson <patrick.eriksson (at) chalmers.se>

The ARTS program is free software; you can redistribute it
and/or modify it under the terms of the GNU General Public
License as published by the Free Software Foundation; either
version 2, or (at your option) any later version.

This program is distributed in the hope that it will be
useful, but WITHOUT ANY WARRANTY; without even the implied
warranty of MERCHANTABILITY or FITNESS FOR A PARTICULAR
PURPOSE. See the GNU General Public License for more
details. 

You should have received a copy of the GNU General Public
License along with the program; if not, write to the Free
Software Foundation, Inc., 59 Temple Place - Suite 330,
Boston, MA 02111-1307, USA. 
\end{code}



%--- Contributing authors -----------------------------------------------------
%
\newpage
\thispagestyle{plain}
%
\begin{center}
  {\Large \bf Contributing authors}
\end{center}
\vspace*{10mm}
\begin{tabular}{lp{10mm}l}
  \hline
  {\bf Author/email} & & {\bf Main contribution(s)} \\
  \hline
  Stefan Buehler$^a$ & & Editor, Sections \ref{sec:concept},
  \ref{sec:absorption} and \ref{sec:batch}.\\
  sbuehler (at) ltu.se & &        \\
%  \hline
%  Cory Davis$^d$ & & Section FIXME. \\
%  cory.davis (at) metservice.com & & \\
 \hline
  Claudia Emde$^c$ & & Sections \ref{sec:clouds} and \ref{sec:scattering}.\\
  claudia.emde (at) dlr.de & & \\
  \hline
  Patrick Eriksson$^b$ &  & Editor, 
  Sections \ref{sec:atmosphere}, \ref{sec:rte}, \ref{sec:surface} , 
  \ref{sec:ppath}, \ref{sec:winds} and \ref{sec:wfuns}.\\  
  patrick.eriksson (at) chalmers.se & & \\
  \hline
  Oliver Lemke$^a$ & & Latex fixes.\\
  olemke (at) core-dump.info & & \\
  \hline
 &&\\
\end{tabular}


\noindent
The present address is given for active contributors, while for others
the address to the institute where the work was performed is given:\\
$^a$ Department of Computer Science, Electrical and Space Engineering,
Division of Space Technology, Lule{\aa} University of Technology, Box
812, SE-98128 Kiruna, Sweden. \\
$^b$ Department of Earth and Space Science, Chalmers University of Technology,
SE-41296 Gothenburg, Sweden. \\
$^c$ Institute of Environmental Physics, University of Bremen, P.O. Box 33044, 
D-28334 Bremen, Germany. \\
%$^d$ Institute for Atmospheric and Environmental Science, University of 
%Edinburgh, EH93JZ Edinburgh, Scotland, UK. \\


%--- Create an empty page
%
\newpage
\thispagestyle{empty}
\rule{0pt}{10pt}
\newpage

\pagenumbering{roman}
\tableofcontents

\cleardoublepage
\pagenumbering{arabic}
     

% ===========================================================================
% === The chapters
%============================================================================


\include{concept}

\include{atmosphere}
\include{rte}
\include{absorption}
\include{surface}
\include{ppath}
\include{winds}
\include{batch}
\include{clouds}
\include{scattering}
\include{wfuns}
\include{sensor}
%\include{reduction}
%\include{wfuns_atm}
%\include{measerr}
%\include{wfuns_sensor} 




\part{Bibliography and Appendices}
%
\bibliography{references}


\part{Index}
%
\printindex


%===   End of report   =====================================================
\end{document}


%%% Local Variables: 
%%% mode: latex
%%% TeX-master: t
%%% End: 
